nkCENWP-OP-B                                                                                                                          04 December, 2006
MEMORANDUM FOR CENWP-OP, ATTN: Randy Bailey

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 48 of 2006, which covers the period from 25 November through 02 December 2006.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 130.2 to 161.1 Kcfs.  Daily average project tailwater elevation ranged from 12.7’ to 14.9’ msl.  Daily average powerhouse forebay elevation ranged from 73.6’ to 74.2’ msl.  Daily average river temperature ranged between 47( and 50(F.  Secchi disk measurements averaged 2.5 to 4.0 feet.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill averaged 1.0 to 5.4 kcfs.  
1.3 Unit Operation.  
1.3.1 Unit 2 remains out of service for turbine rehab.
1.3.2 Unit 8 remains out of service for turbine rehab. 

1.3.3 Unit 10 remains out of service for turbine rehab.

1.3.4 Unit 13 remains out of service for a four year overhaul.

1.3.5 Unit 16 remains out of service for VBS modifications.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby, as necessary, during nighttime hours to float debris. 

1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  The following AWS valves were closed.
2.1.1 FV5-9- from 0708 until 0740 on 27 November for cleaning.
2.2 STS/VBS Inspections.  STSs/VBSs were inspected this reporting week.
	Unit
	Date
	Run Hours
	Comments

	11
	11/29
	828
	STS-A replaced with spare STS

	12
	11/29
	857
	

	13
	
	
	OOS

	14
	11/29
	607
	

	15
	11/29
	859
	STS-A and B unavailable due to VBS mods in U16

	16
	
	
	OOS

	17
	11/29
	858
	

	18
	11/29
	865
	


Table 1.  STS and VBS inspections.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage this week.
3 RESEARCH
3.1 ERDC TDG.  Sensors remain installed in the tailrace.

3.2 NOAA Fisheries- Gilbreath.  Fish continue to be held.  
3.3 USGS lamprey.  Lamprey held in the FERL for tagging evaluation.
3.4 WDFW Fish Counting.  Video counting at all three ladders.
4 FISHWAYS 

4.1 Project biologists inspected 26 through 30 November 2006.
4.2  Fish Passage Plan Observations.
	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR#

	WASHINGTON SHORE

	11/26-27, 11/29-30
	Weir 37
	Reading >1.0’
	Unknown
	No action taken
	

	11/26
	UMT weir
	Reading <1.0’
	Trashracks full
	Riggers will clean on Monday.
	

	11/30
	Weir 38
	Reading <1.0’
	Unknown
	No action taken
	

	CASCADES ISLAND

	11/26, 11/29
	Cascades Is
	Head at trashracks
	High debris loads
	Riggers will clean on Monday.
	

	
	
	
	
	
	

	BRADFORD ISLAND AND BRANCHES

	11/26-30
	PH1 WG 64
	Submergence incorrect
	Gate keeps tripping.
	Electricians to repair on Monday.
	


Table 2.  FPP Out of Criteria items.

4.2.1 PH2 electronic velocity meter was incorrectly installed.  Will be corrected this winter.
4.2.2 CI FG6-12 is closed due to a bent shaft.  OOC when TW exceeds 13’.  Will be repaired this winter.  
4.2.3 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.

4.2.4 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 07/08.  

4.2.5 FG3-3 stuck partially open.  Repairs are scheduled for winter 07/08.  TR#07-2660

4.3 Fishway Temperatures.  Measurements are taken three weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.   
	Date
	WS average
	BI average

	11/26
	50.4
	50.1

	11/27
	49.6
	49.4

	11/28
	49.3
	49.0

	11/29
	48.1
	48.0

	11/30
	47.6
	47.4


Table 3.  Bonneville Fishway temperatures.
4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration is checked on the weekends.  Cascades Island entrance is significantly out of calibration.
4.5.2 Cascades Island and B-Branch PLCs are in manual.  
4.5.3 Stab plates remain installed in the PH2 FOGs.
4.5.4 Cascades Island exit remains open.
4.5.5 Due to the UMT closure, Weir 67 is reading greater than 1.0’ +/- .1’.  
4.6 Juvenile Fishways.
4.6.1 DSM1 will continue to remain OOS until removal in winter 07/08.
4.6.2 There is an adult salmonid in DSM2 Add-in channel.  The screen and add-in bars are in proper position.  Biologists will attempt rescue whenever they are on inspection and able to see the fish.

4.6.3 Due to a lack of PLC control, DSM2 orifice F1 B-N is in local and open as of 2222 on 22 November.

4.7 Lamprey Fishways.

4.7.1 The Bradford Island lamprey ramp was winterized on 21 November.
5 CONSTRUCTION
5.1 B1 ITS Wall Removal.  Plans are to remove the wall, slope the floor, and install three automated gates.
5.2 PH1 STS/VBS removal and TIES removal.  Six TIES and 32 PH1 STSs/VBSs will be scrapped.

5.3 PH2 VBS mods.  Unit 16 is OOS.  Unit 11 will go OOS the first week of December and 15 will follow later.
6 HAZMAT, SPILLS AND CLEANUP – nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 22 November replaces the tailwater operation outlined in the 20 November teletype (which modified the 17 November teletype).  Effective until further notice, but not during any short term chum study operation, a hard tailwater constraint of a minimum 11.8’ during all hours; a hard tailwater constraint of 11.8’ to 12.3’ (targeting 12.0’) during all hours with a priority between 0700 until 1900, except during pulses; at least one pulse of water to last at least 20 minutes and provide at least 13.5’ tailwater.  If there is adequate water to provide two pulses, the teletype requests they occur several hours before dawn and several hours after dusk, if there is only enough water for one pulse, it should occur around midnight.  This operation will be superseded only by short term chum study operations scheduled by USGS.  The intent of this request is to facilitate chum spawning at lower elevations, yet still keep redds at higher elevations wet.  
7.2 A teletype sent on 27 November detailed the chum spawning studies planned for 28 and 30 November.  Tests are scheduled to occur during the evening and will require ramping flows up, holding the target for eight hours then ramping back down.  During testing the tailwater elevations below override the elevations outlined in the teletype sent on 22 November (7.1).  If 16.0’ can not be reached, hold tailwater as high as possible.

7.2.1 28 November 1500- increase from 12.0’ to 14.0’

7.2.2 28 November 1600- increase from 14.0’ to 16.0’

7.2.3 29 November 0100- decrease from 16.0’ to 14.0’

7.2.4 29 November 0200- decrease from  14.0’ to 12.0’

7.2.5 30 November 1500- increase from 12.0’ to 14.0’

7.2.6 30 November 1600- increase from 14.0’ to 16.0’

7.2.7 01 December 0100- decrease from 16.0’ to 14.0’

7.2.8 01 December 0200- decrease from  14.0’ to 12.0’

7.3 A teletype sent on 30 November requests a steady tailwater elevation from 1000 until 1300 on 01 December to assist with survey work determining the elevation of chum redds near Ives Island.  

7.4 A teletype sent on 28 November requests a forebay soft constraint of 71.5’ – 72.5’ from 0700 until 1600 (or as long as possible) on 04 December.  This operation is to facilitate environmental clean up at lower elevations near Hood River.

7.5 A teletype sent on 01 December grants permission to operate Unit 16 in normal priority without fish screens in place.  All fish screens will be pulled starting 18 December.

JAMES R. MAHAR, P.E.

Operations Manager
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